
Morning status 6 Jan
Status 1/6/2011: Oncley&Dong onsite.

Fog this morning, about 0.2mi visibility.  Frost on van windows at the Rez.

Stn1 krypton frosted up immediately in the evening last night.

Stn3 krypton also went to 0 early this morning

Stn4 krypton appears to be failing just now

Our plan:

Stn1 quick visit to defrost sensors and to shoot prop angle (might not "stick" since lots of fog still present)

Stn6 normal visit

Stn7 normal visit and add SPN-1 and klean krypton if time

Return to campus for 1:00 meeting
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